



# 

**Citation:** Okello WO, Okello AL, Inthavong P, Tiemann T, Phengsivalouk A, Devleesschauwer B, et al. (2018) Improved methods to capture the total societal benefits of zoonotic disease control: Demonstrating the cost-effectiveness of an integrated control programme for *Taenia solium*, soil transmitted helminths and classical swine fever in northern Lao PDR. PLoS Negl Trop Dis 12(9): e0006782. https://doi.org/10.1371/journal. pntd.0006782

**Editor:** Agnes Fleury, Instituto de Investigaciones Biomedicas, UNAM /Instituto de Neurologia y Neurocirugía, MEXICO

Received: June 20, 2018

Accepted: August 23, 2018

Published: September 19, 2018

**Copyright:** © 2018 Okello et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** This research was funded by the Australian Centre for International Agricultural Research (ACIAR) Animal Health Programme, RESEARCH ARTICLE

Improved methods to capture the total societal benefits of zoonotic disease control: Demonstrating the cost-effectiveness of an integrated control programme for *Taenia solium*, soil transmitted helminths and classical swine fever in northern Lao PDR

# Walter O. Okello<sup>1\*</sup>, Anna L. Okello<sup>2</sup>, Phouth Inthavong<sup>3</sup>, Tassilo Tiemann<sup>4</sup>, Ammaly Phengsivalouk<sup>3</sup>, Brecht Devleesschauwer<sup>5,6</sup>, Alexandra Shaw<sup>1,7</sup>, John Allen<sup>2</sup>

1 Division of Infection and Pathway Medicine, University of Edinburgh, Edinburgh, United Kingdom,

3 Department of Livestock and Fisheries, National Animal Health Laboratory (NAHL), Vientiane, Lao PDR,

**4** Tropical Forages Program, Centro Internacional de Agricultura Tropical (CIAT in Asia), Vientiane, Lao PDR, **5** Department of Epidemiology and Public Health, Sciensano, Brussels, Belgium, **6** Department of Veterinary Public Health and Food Safety, Faculty of Veterinary Medicine, Ghent University, Merelbeke, Belgium, **7** A P Consultants, Andover, United Kingdom

\* walterokell@gmail.com

# Abstract

# Background

Control and elimination of zoonotic diseases requires robust information about their effect on both human and livestock health in order to enable policy formulation and the allocation of resources. This study aimed to evaluate the cost-effectiveness of controlling *Taenia solium* taeniasis/cysticercosis in both humans and pigs, and soil-transmitted helminths (STH) in humans by integrating their control to on-going human and animal health control programmes in northern Lao People's Democratic Republic.

# Method

A cross-sectional study was carried out in 49 households, focusing on the prevalence of *T*. *solium* taenias/cysticercosis and soil transmitted helminths before and after a twelve month intervention. The village data was collected using a semi-structured questionnaire through a door-to-door survey. The village data was then projected to the wider northern Lao PDR population using stochastic modelling and cost-effectiveness ratio (after aggregating the net cost to capture both human and animal health parameters) and GDP per capita as a threshold, to determine the cost-effectiveness of the integrated control of *T*. *solium* taeniasis/ cysticercosis and STH, assuming linear scaling out of the intervention. The zoonotic DALY (zDALY) approach was also used as an alternative method of estimating the cost-effectiveness ratio of controlling *T*. *solium* taeniasis/cysticercosis in humans and pigs.

<sup>2</sup> Australian Animal Health Laboratory (AAHL) Regional Program, CSIRO, Geelong, VIC, Australia,

grant AH2009/001 and AH/2009/019. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** AP consultants was not involved in any way in the actual implementation of the project nor does it produce any commercial products nor did it fund any part of the project and none of the co-authors are employed or affiliated with this company except Dr Alexandra Shaw who declares no competing interests.

## Findings

Using cost-effectiveness analysis after aggregating the net cost and control of *T. solium* taeniasis/cysticercosis alone as the base case, the study found that simultaneous control of *T. solium* taeniasis/cysticercosis in humans and pigs, STH in humans and Classical Swine Fever (CSF) in pigs was USD 14 per DALY averted and USD 234 per zDALY averted using zDALY method hence considered highly cost-effective whereas controlling *T. solium* taeniasis/cysticercosis without incorporating STH and CSF was the least cost-effective (USD 3,672 per DALY averted). Additionally, the cost-effectiveness of controlling *T. solium* taeniasis/cysticercosis in people and pigs using zDALY as an alternative method was USD 3,662 per zDALY averted which was quite close to our findings using the aggregate net cost method.

# Conclusion

The study showed that control of *T. solium* taeniasis/cysticercosis alone in humans and pigs is not cost-effective in northern Lao PDR whereas control of STH is. Consequently, integrating *T. solium* taeniasis/cysticercosis control with other cost-effective programmes such as STH and CSF markedly improved the cost-effectiveness of the intervention. This is especially important in low resource countries where control of zoonotic neglected tropical diseases could be integrated with the human and animal health sectors to optimize use of the limited resources.

# **Trial registration**

Australia New Zealand Clinical Trials Registry (ANZCTR) ACTRN12614001067662.

# Author summary

A study was conducted in northern Lao PDR to ascertain the cost-effectiveness of controlling Taenia solium (T. solium taeniasis/cysticercosis) using five approaches namely: i) T. solium taeniasis/cysticercosis alone in the human population (the base comparator), ii) T. solium taeniasis/cysticercosis and soil transmitted helminths (STH) in the human population, iii) T. solium taeniasis/cysticercosis alone in the human and pig population, iv) T. solium taeniasis/cysticercosis in the pig population and STH in humans, and v) T. solium taeniasis/cysticercosis, STH and Classical Swine Fever (CSF) in humans and pigs. Using cost-effectiveness ratio (after aggregating the net cost and using zDALY approach as an alternative method), the study found that the simultaneous control of T. solium taeniasis/ cysticercosis, STH and CSF in human and pig population was USD 14 per DALY averted and USD 234 per zDALY averted thus considered highly cost-effective whereas control of T. solium taeniasis/cysticercosis alone in the human and pig population was the least costeffective as it was found to be USD 3,672 per DALY averted using the aggregate net cost method and USD 3,662 using the zDALY approach,. We concluded that inclusion of STH and CSF to T. solium taeniasis/cysticercosis mitigation efforts drastically improved the overall cost-effectiveness of the intervention in northern Laos where all the three diseases are endemic.

#### Introduction

*Taenia solium* taeniasis-cysticercosis complex is a zoonotic Neglected Tropical Disease (zNTD) found throughout many parts of Asia, Africa and Latin America, particularly where pigs and humans co-exist in areas of poor sanitation and hygiene [1–2]. Being the most important food-borne parasite worldwide and ranked fourth among all food-borne pathogens [3], there is a growing requirement for improved understanding of the global burden and demonstration that control is cost-effective [4].

The World Health Organization (WHO) has promoted a scale up of T. solium taeniasis/cysticercosis control and elimination efforts by 2020, buoyed by its status as one of six diseases identified as 'potentially eradicable' by the International Task Force for Disease Eradication [5]. Amongst other things, the task force recommends integrated control strategies, consideration of economic factors and assessment of the impact of mass drug treatment on co-endemic parasitic diseases such as soil-transmitted helminths (STH) to help promote support for eradication [5]. Following this, there is broad consensus that the economic analysis of zoonoses control programmes should be based on a holistic measurement of 'total societal benefits' as compared to simply calculating the total costs of controlling disease in humans and in animal reservoirs [6]. This requires an understanding of the level of integration [7-8] and comprehensive economics metrics to compare cost-effective control methods [9]. In the past, integrated control of neglected tropical diseases such as trachoma and primary healthcare [10], schistosomiasis and STH using common drugs [11], rabies in the animal health sector [12] among others have been attempted with varying results. Although, a policy of integrated control of neglected tropical diseases is highly beneficial [13-15], studies on the cost-effectiveness of such an approach are rare [16].

This study aims to quantify the overall cost-effectiveness of a successful 'rapid impact' T. solium taeniasis/cysticercosis control programme that treated both pigs and humans, resulting in a significant (p < 0.001) T. solium taeniasis/cysticercosis reduction of 77.4% over a sixteenmonth period in a smallholder farming system in Southeast Asia typically characterized by reliance on family labour, small farm size, minimal input and low income [17]. Apart from the T. solium taeniasis/cysticercosis control program in Laos, other studies have shown that the prevalence of epilepsy in Lao PDR is 7.7 per 1,000 people [18] with high fatality rates [19–20] and low levels of healthcare [21–22]. To date, whilst models have suggested that a combined therapeutic approach in both pig and human hosts will result in the greatest sustained impact on parasite levels [23], few research interventions have explored this concept in practice [24-25]. In this study we also evaluated combined human mass drug administration (MDA), oxfendazole deworming of pigs and vaccination of pigs using TSOL18 and Classical Swine Fever (CSF) vaccines based on the holistic One Health intervention undertaken by Okello et al. (2016) [17] which aimed to optimize its total societal value through integration into existing district-level programmes driven by the Lao PDR Ministries of Health and Agriculture in the target area and carrying out joint activities. On the human side, this was achieved through two rounds of community mass drug administration (MDA) with a three day albendazole 400mg protocol to decrease the level of environmental contamination with tapeworm eggs and incorporate STH control and thus align with the Ministry of Health's ongoing STH control objectives [26–27]. During the MDA, local government medical staff visited all participating households for five consecutive days administering drugs, monitoring for adverse reactions and carrying out screening exercises for epilepsy. The human health intervention excluded pregnant women and children less than six years old. The standard porcine anti-cysticercosis intervention (which excluded pregnant or lactating pigs as well as those earmarked for sale), consisting of TSOL18 vaccination and oxfendazole (OFZ) at 30mg/kg [28-29], also

incorporated Classical Swine Fever (CSF) vaccination, an important porcine production-limiting disease in Southeast Asia [30] and a priority disease for the Lao PDR Ministry of Agriculture. Classical Swine Fever, especially genotype 2.2, is endemic in northern Lao PDR and it is characterized by abortions and stillbirths of sows, as well as lack of appetite, anorexia, and high-mortality (can reach 100%) among nursery pigs [31-32]. It is hoped that this methodology and findings will help drive similar cost analyses for T. solium taeniasis/cysticercosis and other zNTD interventions, whilst simultaneously encouraging the consideration and inclusion of possible collateral benefits into control of other diseases under a true One Health approach. Consequently, to make this study have a wider applicability, a research question and null hypothesis were developed. The research question was 'how does the inclusion of STH and CSF affect the cost-effectiveness of T. solium taeniasis/cysticercosis control?', and based on this question, the null hypothesis was that inclusion of STH and CSF has no significant impact on the cost-effectiveness of T. solium taeniasis/cysticercosis control. The base case was the T. solium taeniasis/cysticercosis control alone without inclusion of STH and CSF while the comparators were the T. solium taeniasis/cysticercosis control strategies that included STH and CSF.

# Methodology

#### Study area

The study was conducted in Mai district, Phongsaly province in the northern region of Lao PDR. The target area consisted of a homogenous Tai Dam population of around 400 people in 55 households, where the pre-intervention *T. solium* taeniasis/cysticercosis prevalence was found to be one of the highest recorded globally to date [33]. The Tai Dam are an ethnic group from northern Lao PDR, Vietnam, Thailand and China with strong cultural ties to animal sacrifice, using pigs, chickens and buffalo during various ceremonies and festivities that connect them with a higher spiritual world [34].

# Data collection

After seeking a written consent and conducting a door to door household census, a semi-structured questionnaire was used to determine household characteristics, pig productivity and human health parameters; including reporting on epilepsy through screening [35–36] in 49/55 (89.1%) of village households. The initial baseline survey, conducted in October 2014, included a 12 month recall for livestock productivity data regarding pig production. During the subsequent 18-month intervention [17], economic monitoring occurred via every six months updates on changes in the village pig population (births, deaths, sales, purchases etc), human health parameters, and response to both the human and pig interventions which were concurrently undertaken.

#### Economic evaluation of the intervention costs and benefits

The total societal view, where all resources are captured irrespective of who incurred or benefited from them, was central to cost computation in this study. The intervention strategies being compared in this study were: i) *T. solium* taeniasis/cysticercosis alone in the human population (the base case), ii) *T. solium* taeniasis/cysticercosis and soil transmitted helminths (STH) in the human population, iii) *T. solium* taeniasis/cysticercosis alone in the human and pig population, iv) *T. solium* cysticercosis in the pig population and STH in humans, and v) *T. solium* taeniasis/cysticercosis, STH and Classical Swine Fever (CSF) in humans and pigs. These interventions represented all the possible scenarios public health policy makers would

face in regards to control of T. solium taeniasis/cysticercosis in Laos; intervention strategies two to five were the comparators. The questionnaire captured both monetary and time expenditures borne by village inhabitants (private costs) resulting from symptoms or disease associated with T. solium taeniasis/cysticercosis or STHs (direct costs of health seeking treatment). The questionnaire also captured private costs incurred by smallholder farmers from pig rearing, via gross margin analysis of the pig enterprise in the target area. Public (project) costs were allocated to either the human or pig cost centres using a micro-costing approach [37], enabling their analysis as a constituent of the overall project cost without double counting. Capital depreciation, which was the only capital cost, was estimated using the straight line method [38] and aggregated amongst the cost centres. Examples of the human intervention project cost centre included the cost of albendazole tablets, capital depreciation and logistical costs. The project costs incurred from the pig intervention included the cost of oxfendazole, TSOL18 and CSF vaccine, plus other recurrent expenditures. Also, secondary data such as cost of treatment and drugs were fitted to gamma distribution using the fitdistrplus package for R [39] and analysed using a Monte Carlo simulation to estimate the 95% uncertainty interval. For the purposes of analysis, the costs and benefits were divided into human (non-monetary and monetary) and pig (monetary), although execution of both interventions was combined.

#### Calculating the total human benefit of the intervention

DALYs represent the non-monetary human disease burden, calculated through combining the years of life lost due to premature death (YLL) and years lived with disability (YLD) [40–41]. The epidemiological parameters used for the DALY calculations of neurocysticercosis (NCC) and STH were obtained using a combination of empirical data derived from household questionnaires and secondary literature sources inputted into R software (version 3.2.2) [42]. Preference was given to secondary data obtained from the study area or in other districts of Laos. However data from south-east Asia was used in cases where there was no information available in the study area or other parts of Laos.

Since the accuracy of DALY estimates rely heavily on the information obtained for its computation, secondary data were fitted to uniform and beta distribution using the fitdistrplus package for R [43] and analysed using a Monte Carlo simulation, allowing for estimation of uncertainty to the DALY estimate [44]. Also, the discount rate and social weighting (K and r values in the YLL equation) were set at zero to allow for comparison with other studies and the burden of *T. solium* taeniasis/cysticercosis and STH averted was represented as DALY [0, 0, 0] [40].

A door-to-door survey [45] was undertaken to estimate the number of epilepsy cases in the target area, with the prevalence converted to incidence by dividing it by illness duration [46]. The proportion of epilepsy due to NCC was estimated using secondary data, given the study did not diagnostically confirm reported epilepsy cases. The estimated STH prevalence within the target area [27] was combined with STH prevalence data from other northern Lao PDR provinces [47–48], then converted to incidence levels [48–49]. Tables 1 and 2 provide a summary of all epidemiological parameters that were used to estimate the non-monetary burden of *T. solium* taeniasis/cysticercosis and STH in the broader northern Lao PDR population.

# Calculating the total livestock benefit of the intervention

The animal arm of the zoonotic disease burden is represented in this case by pig livestock production losses, incorporating the costs of both livestock death and morbidity such as lowered fecundity, weight loss leading to a reduced sale price, or carcass condemnation due to the presence of cysts. Losses to the pig production enterprise were determined via a 'livestock

Parameter	Value	Distribution	Data source
Population	1,141, 785	Fixed	[50]
Prevalence of epilepsy	7.7 per 1,000	Fixed	[18]
Proportion of epilepsy due to NCC	0.07	Uniform (0.002-0.158)	[51-53]
Case fatality ratio for NCC (%)	0.11	Beta (0.02–0.22)	[18]
Age of onset for age group 0–4 years	2.0	Fixed	[54]
Age of onset for age group 5–14 years	9.9	Fixed	[54]
Age of onset for age group 15–44 years	26.9	Fixed	[54]
Age of onset for age group 45–59 years	51.9	Fixed	[54]
Age of onset for age group over 60 years	73.6	Fixed	[54]
Average duration for age group 0–4 years	1.4	Fixed	[55]
Average duration for age group 5–14 years	2.0	Fixed	[55]
Average duration for age group 15–44 years	3.6	Fixed	[55]
Average duration for age group 45–59 years	2.8	Fixed	[55]
Average duration for age group over 60 years	1.6	Fixed	[55]
Disability weights for epilepsy treated	0.076	Uniform (0.047-0.106)	[56]
Disability weights for epilepsy untreated	0.424	Uniform (0.279–0.572)	[56]

Table 1.	Epidemiological	parameters used	l for computatio	n of the burden of NO	CC

https://doi.org/10.1371/journal.pntd.0006782.t001

production' section of the household questionnaire which evaluated the numbers of pigs bought (including the prices they were bought), sold (including prices they were sold), died (including reasons for the death) and born per household over the given time period. A second element considered the private (borne by livestock keepers) and public (project) animal health expenditure in terms of both time and money, expressed as a component of the variable costs. The gross margin (expressed as the net benefit) was then calculated to determine the change in household income pre and post intervention according to the standard formula: Gross margin = [livestock output]–[variable cost] [60], where livestock output is defined as = [(animals and produce 'out')–(animals and produce 'in')] plus change in herd value. The change in herd value is expressed as [closing valuation (the total number of pigs at the end of the year multiplied by their value)—opening valuation (the total number of pigs at the beginning of the year multiplied by their value)]. The value of the pig is a function of its weight which is correlated

Table 2. Epidemiological parameters used for computation of the burden of STH.

Parameter	Value	Distribution	Data source
Prevalence of Hookworm	0.46	Beta (0.45–0.47)	[27, 47–48]
Prevalence of Ascaris lumbricoides	0.12	Beta (0.12–0.13)	[27, 47-48]
Prevalence of Trichuris trichura	0.4	Beta (0.39–0.42)	[27, 47]
Case fatality ratio for STH	0.0014-0.08	Beta (0.000–0.001)	[57]
Age of onset for age group 0–4 years	2	Fixed	[58]
Age of onset for age group 5–14 years	10	Fixed	[58]
Age of onset for age group 15–44 years	30	Fixed	[58]
Age of onset for age group 45–59 years	50	Fixed	[58]
Age of onset for age group over 60 years	70	Fixed	[58]
STH duration	1	Fixed	[58]
Disability weight Ascaris Lumbricoides	0.067	Uniform (0.0108-0.1245)	[57, 59]
Disability weight trichuriasis	0.067	Uniform (0.0108–0.1245)	[57, 59]
Disability weight hook worm	0.063	Uniform(0.0041-0.1245)	[57, 59]

https://doi.org/10.1371/journal.pntd.0006782.t002

with its age and health status; a pig's weight is influential in determining its selling or buying price. Typically younger pigs (piglets and weaners) cost less to buy and sell compared to older pigs (growers, sows and boars). The value of the pigs which was not captured by the question-naire was obtained from key informant interviews (which composed of 12 farmers, 9 traders, 3 animal health technicians and 2 veterinarians) by determining the most probable, minimum and maximum selling of each pig type and then using beta-PERT distribution in r software (mc2d r package) to determine the mean selling price in a smooth parametric distribution [42]. Variable costs include the costs of pig rearing incurred by the farmer plus any expenses of project participation (for example repair of pig pens).

#### Projection of findings to the larger northern Lao PDR region

The cost-benefit analysis was projected to the broader northern Lao population over the course of one year in order to estimate the cost-benefit of control at a regional level that would more accurately reflect future control programmes assuming linear scaling. The total human population in the four Northern provinces considered for this projection (Phongsaly, Huaphan, Luang Prabang and Oudomxay) was 1,141,785; comprising 572,211 females (0–4 years age group had 71,194 females, 5–14 years age group had 150,213 females, 15–44 years age group 261,812 females, 45–59 years age group had 54,544 females and over 60 years age group had 34,448 females) and 569,574 (0–4 years age group had 71,766 males, 5–14 years age group had 154,355 males, 15–44 years age group had 258,017 males, 45–59 years age group had 53,540 males and over 60 age group had 31,896 males) males [50].

The number of pig rearing households, total number of pigs in each province and the average number of pigs per household in each of the four provinces was obtained from the Lao agricultural census [61]. In the four northern Lao provinces there were a total of 104,700 pig rearing households and the total number of pigs was 351,200. Computation of the gross margin per household entailed adjusting the income from the pig enterprise depending on the mean number of pigs per province, since the gross margin is highly dependent on the herd size.

# Analysing the total societal benefit of the intervention in relation to its total cost

The overall capacity of a public health intervention to prevent unwanted human health outcomes (such as mortality and prolonged morbidity resulting from disease presence) may be indicated by the number of DALYs averted [62]. The total cost-effectiveness of this *T. solium* taeniasis/cysticercosis control programme, in relation to the total costs and benefits accrued in both the human and pig arms of the intervention, were evaluated using cost-effectiveness ratio a standard measure of cost utility analyses [63]. However, since the study had both costs from the agricultural and health sectors, the aggregated net cost was calculated by subtracting the reduced human health cost (i.e. decrease in expenditure incurred by medical services, patients and their households), reduced animal health losses (i.e. mortality, weight etc.), reduced animal health expenditure (i.e. decrease in expenditure incurred by veterinary services and farmers) from the project cost as shown in Eq 1 [61].

NC/DALY averted = 
$$(CI - RE_{H} - RL_{A})/RB_{H}$$
 (1)

Where NC/DALY averted is the net cost per DALY averted for each intervention, CI is the total project cost (in USD),  $RE_H$  is the reduced human health expenditure (in USD),  $RL_A$  is the reduced animal health losses, and  $RB_H$  is the DALY averted. The zoonotic DALY (zDALY) was also used as an alternative approach to estimate the cost-effectiveness ratio of controlling

*T. solium* taeniasis/cysticercosis [64–65]. Just like the aggregate net cost method, the zDALY also provides a framework for combining the human burden and the losses incurred by live-stock keepers in single metric and it does this by considering the monetary impact on livestock keepers in terms of a time trade-off, in the sense of the value of people's time required to recoup these losses, and by using gross domestic product (GDP) as a numéraire, it convert these into a non-monetary time unit called 'animal loss equivalent' (ALE) [65]. Equally, the reduced expenditure on health due to an intervention is converted into 'health loss equivalent' (HLE). Another method for analysing total societal benefit of controlling zoonotic diseases is the separable cost method [16]. Although the main method used in this study was aggregated net cost, zDALY approach was also used to; i) capture the cost of zDALY averted for *T. solium* taeniasis/cysticercosis, and ii) costs per zDALY averted across all diseases rather than just for *T solium* utilising the ALE component of CSF.

To estimate the cost-effectiveness of controlling *T. solium* taeniasis/cysticercosis, the study applied the WHO cost-effectiveness thresholds, which considers an intervention as 'highly cost-effective' if the cost per DALY averted is less than the country's GDP per capita; 'cost-effective' if the cost per DALY is between one and three times the GDP per capita; and 'not cost-effective' if the cost per DALY exceed three times the country's GDP per capita [66]. In Lao PDR the GDP per capita at the time of the study was USD 1,793 [67]. Nominal current (year 2015 without adjusting for price inflation) market prices were used throughout for both disease control costs and livestock production costs and returns, thus reflecting the realities facing the health and veterinary services, pig producers and human patients in Lao. Accordingly the GDP value selected was also the nominal, or Atlas value, rather than being in international dollars adjusted for purchasing power parity.

#### **Ethics statement**

Ethical approval for this study was granted by the Lao PDR Council of Medical Science National Ethics Committee for Health Research (NECHR), approval number 013/NECHR, and Australia's Commonwealth Scientific and Industrial Research Organisation (CSIRO) Animal, Food and Health Sciences Human Research Ethics Committee (CAFHS HREC), approval number 13/10. The study was registered with the Australia New Zealand Clinical Trials Registry (ANZCTR), trial number ACTRN12614001067662.

# Results

#### Descriptive socioeconomic characteristics

From a total of 55 target area households, 49 (89%) were included in the study; six households had relocated during the course of the study, hence were not included in the final calculations. The total number of people in the 49 households was 375 comprising 178 males (0–4 years age group had 20 males, 5–14 years age group had 45 males, 15–44 years age group 85 males, 45–59 years age group had 18 males and over 60 years age group had 10 males) and 197 females (0–4 years age group had 86 females, 45–59 years age group had 18 females and over 60 age group had 11 females). According to information obtained from key informant interviews, all pigs were left to roam freely, with the larger ones penned during the rice harvesting season. Questionnaires indicated the average weight of sold pigs was around 23 kilograms (kg) and the price of pork around USD 3/kilogram. Out of the 49 households, 41 (83%) did not have toilets, whereas 8 (16%) had toilets and used them.

The baseline pig population in the target area was 270, with a mean number of 5.2 (standard deviation 4.9) pigs per household. The herd structure consisted of 28 boars (10%), 64 sows

(24%), 53 weaners (20%), 32 growers (12%) and 93 (34%) piglets. Post intervention, pig numbers had increased by 53% to 414, with a mean number of 8.4 (SD 6.1) pigs per household. Changes in the herd structure, particularly in the grower category, were evident with 27 boars (7%), 74 sows (18%), 34 weaners (8%), 182 growers (44%) and 97 piglets (23%) post intervention (Fig 1). Before intervention, mortality was 48.3% (185/386) comprising of 138 piglets, 32 weaners, 7 growers, 5 sows and 3 boars. After the intervention the mortality dropped to 8.4% (37/440) comprising of 16 piglets, 9 weaners, 6 growers, 3 sows and 3 boars. There were 386 responses on the reasons for death and these were mentioned as follows; diseases (69%), lack of feed (13%), accidents (1%), dog bites (2%), poisoning (1%), gunshot wounds (1%), and piglets dying from low milk supply from sow during lactation (1%) and still birth (12%).

#### Project costs for the combined human and pig intervention

**Costs (capital and recurring) incurred by the project.** During the first two rounds of control, in October 2014 and March 2015, the human and pig interventions were undertaken simultaneously. This resulted in combined project expenditures for these activities, with the final October 2015 pig intervention treated as a single activity. The only capital cost was for vehicle depreciation, while the recurrent costs included drugs, vaccines, fuel and staff costs. The total capital and recurrent expenditure for the human cost centre was USD 3,498, and the pig cost centre USD 21,085. The pig cost centre was further divided into TSOL18, OFZ and CSF vaccination.

**Direct financial costs incurred by the project.** The human arm of the intervention consisted of two rounds of MDA with an albendazole 400 mg triple dose protocol and one monitoring exercise, as described by Ash et al (2015) [27]. The direct financial project costs for these activities were the cost of albendazole tablets (1,048 tablets at USD 0.1 each) delivered to 375 human participants. The pig intervention (TSOL18 plus OFZ plus CSF) was administered four times over the intervention period (three vaccination activities and one monitoring exercise), with direct financial costs including the cost of CSF vaccine and OFZ. The CSF vaccine was bought in bulk from the government veterinary laboratory based in Vientiane, at a total cost of USD 100 (833 doses at USD 1.2 per 10 doses), with a total number of 828 pigs vaccinated during one year of the intervention. Although in this case the TSOL18 vaccine was donated (MWL Lightowlers, University of Melbourne), for the purposes of this analysis the cost has been included as the expected wholesale commercial cost of around 0.50 USD/dose, 95% uncertainty interval (UI: USD 0.4–0.6) (personal communication Galvmed 2017). The cost of the various component interventions have been summarised in Table 3.



Fig 1. Pre and post intervention pig numbers in the target area.

https://doi.org/10.1371/journal.pntd.0006782.g001

Cost item	Human intervention (USD)	Pig intervention (USD)			Total cost (USD)	
	Mass drug administration (MDA)	TSOL18 vaccination	Deworming with oxfendazole	Classical Swine Fever vaccination	-	
Capital expenditures						
Vehicle depreciation	250	250	250	250	1,000	
Recurrent expenditures						
Value of field staff participation human MDA	1,378	0	0	0	1,378 (95% UI: 1,239– 1,599)	
Value of field staff participation during pig intervention	0	5,052	5,052	5,052	15,156 (95% UI: 12,671–17,791)	
Value of field staff participation during monitoring	148	148	148	148	592 (95% UI: 525– 667)	
Local facilitation fees	75	75	75	75	300 (95% UI: 186– 442)	
Total cost of human drugs	105	0	0	0	105	
Cost of Classical Swine Fever vaccine	0	0	0	100	100	
Cost of TSOL18 vaccination	0	810	0	0	810 (95% UI: 727– 929)	
Cost of oxfendazole dewormer for pigs	0	0	366	0	366	
Additional vehicle hire	315	315	315	315	1,260 (95% UI: 1,184– 1,391)	
Total fuel used	500	500	500	500	2,000 (95% UI: 1,590– 2,619)	
Questionnaire development, pre- testing and printing	38	38	38	38	152 (95% UI: 107– 206)	
Laboratory diagnostics	464	0	0	0	464 (95% UI: 299– 939)	
Miscellaneous	225	225	225	225	900 (95% UI: 818– 1,021)	
Total amount in USD	3,498	7,413	6,969	6,703	24,583	

#### Table 3. Total public (project) costs of the human-pig intervention.

https://doi.org/10.1371/journal.pntd.0006782.t003

# Human health results

**Non-monetary cost of NCC and STH (DALYs).** The clinical manifestations of epilepsy are well known in northern Lao PDR, named '*bah moo*' (crazy pig) in the local language based on the likeness of sufferers to salivating pigs. Door-to-door household surveys in the target area revealed approximately 0.8% of inhabitants self-reported suffering from epilepsy (3/375), based on a clinical description of seizures with or without mouth foaming. Post intervention epilepsy prevalence was reported at 0.2% (1/375); equivalent to 2 people per 1,000. Projecting these values to the broader northern Lao PDR region, the number of people estimated to be suffering from epilepsy pre-intervention was 9,134, resulting in an estimated 2,404 patients with estimated NCC-induced epilepsy in the broader northern region.

Entering these parameters into R software [40], the pre-intervention DALY from NCC was estimated to be 4,958 (95% UI: 4,940–4,975), with a total DALYs of 434 per 100,000 person years. It was assumed that there would be 2,284 people suffering from epilepsy post-intervention; hence the estimated number of people that would suffer from NCC-induced epilepsy in northern Lao PDR was 580. The total DALYs due to NCC was 1,480 (95% UI: 1,475–1,485) post-intervention; and the total DALYs per 100,000 person years due to NCC-induced epilepsy was 130 post-intervention in northern Lao PDR, representing a 70% reduction in the disease burden. The epidemiological parameters of the STH were computed in R (v 3.1.1) after coding,

Disease	DALY Pre-intervention	DALY Post-intervention	DALY averted	Percentage reduction in DALY burden
Trichuriasis	35,250	17,923	17,327	49
Hookworm	37,786	11,168	26,618	70
Ascariasis	17,237	5,104	12,133	70
NCC	4,958	1,480	3,478	70
Total	95,231	35,675	59,556	63

#### Table 4. DALY averted for NCC and STH.

https://doi.org/10.1371/journal.pntd.0006782.t004

yielding the DALY figures pre- and post-intervention for each of the three diseases. Consequently the total DALYs averted was estimated to be 59,556 in northern Lao PDR, representing an overall reduction of 63% of the total disease burden due to NCC and STH as shown in Table 4.

### Patient cost of NCC and STH

**Cost of NCC-associated epilepsy treatment.** Data obtained from the door-to-door survey revealed that only 33% identified persons with epilepsy (PWE) sought healthcare (traditional healer), spending an average of USD 17 (95% UI: 14–19) on treatment per year. Sixty seven percent of PWE did not seek any form of treatment. Based on this, an estimated 793 (33%) NCC-associated epileptic patients sought treatment, at a cost of USD 13,481/year, before the intervention. Post intervention, the number of NCC patients in northern Lao PDR seeking healthcare was estimated to be 191 at a cost of USD 3,247/year, resulting in a post-intervention difference of USD 10,234 in healthcare expenditure as a result of NCC-associated epilepsy.

**Cost of STH treatment.** According to the data obtained from the household surveys using one year recall, 56% (210/375) participants' self-reported symptomatic diarrhoea and anaemia, with 19% of these (40/210) reporting to have sought health care. Of the individuals that sought healthcare, 6 went to the medical doctor, spending an average of USD 0.6 (UI: 0.4– 0.8) each, and the remaining 34 patients went to the village soothsayer, spending an average of USD 0.1 (95%UI: 0.03-0.19) each. Using a gamma distribution, the mean expenditure incurred when seeking healthcare (both from the medical doctor and soothsayer) was USD 0.3 (95% UI: 0.1–0.5). Data from Ash et al. (2015) [27] revealed that the mean STH prevalence in the target area was 53%, consequently an estimated 605,146 individuals were affected by STH, with a total health expenditure of USD 34,493, across the four northern Lao PDR provinces pre-intervention. The estimated post-intervention STH prevalence in the target area was 9.9% [43], projecting to an estimated 113,037 people across the broader region, with a treatment expenditure of USD 6,443. Based on the difference in the amount of money spent on treating NCC and STH within the informal private health sector, the difference in health expenditure after the intervention was USD 28,050. Consequently, the total amount of health expenditure averted was USD 38,284.

# Pig intervention results

**Pigs and pork products 'out'.** The value of pigs and pork products out constituted the total value in USD of all pigs sold, slaughtered or those given as gifts among the inhabitants of the village throughout the intervention. Ninety-six pigs (18 piglets, 10 weaners, 60 growers, 3 sows and 5 boars) were sold at a total value of USD 3,519 in the 12 months leading up to the intervention. Post intervention, this figure had risen to 102 pigs (10 piglets, 5 weaners, 80 growers, 2 sows and 15) sold at total value of USD 8,364. Baseline slaughter numbers (i.e. slaughtered and consumed without condemning) were 21 at a total value of USD 2,232, which

at post intervention had risen to 46 at a total value of USD 3,856. Other pigs transferred out of the village herd (gifts for other reasons) totalled 13, at a total value of USD 768 pre-intervention and largely consisted of younger pigs, increasing to 19 pigs and a value of USD 1,239 post intervention. Therefore total value of pigs sold, slaughtered or gifted out before and after the intervention was USD 6,519 and USD 13,459 respectively; a difference of USD 6,940.

**Pigs and pork products 'in'.** The value of pigs and pork products into the target herd consisted of the sum total value of all pigs purchased or that came into the herd for other reasons, for example received as gifts. In the 12 months leading up to the intervention, a total of 17 pigs (8 sows, 3 weaners and 6 growers) were purchased by the village households, at a total expenditure of USD 960. Post intervention analysis demonstrated that this figure had dropped by the end of the intervention, with 9 pigs purchased (5 sows, 1 weaner, 2 growers and 1 piglet) at a cost of USD 630. Eleven pigs (2 boars, 3 sows, 3 weaners, 1 grower and 2 piglets) were received as gifts pre-intervention, valued at USD 737, decreasing to gifts of 3 pigs (2 sows and 1 grower) worth USD 231 post intervention. Consequently, the overall total decrease in the value of pigs bought or gifted in the village over the intervention period was USD 837.

Gross margin from pig enterprise before and after the intervention. To complete the livestock output calculation, the change in herd value was calculated. This is difference between the closing valuation (total number of pigs at the end of the year multiplied by their value) and the opening valuation (total number of pigs at the beginning of the year multiplied by their value). Using one year recall, the total number of pigs one year before the baseline study (October 2013 to October 2014) was 256 (86 piglets, 50 weaners, 33 growers, 62 sows and 26 boars) valued at USD 12,471 (which represents the opening valuation). During the baseline year there were 270 pigs as mentioned earlier valued at USD 14,372 hence the herd value before the intervention was USD 1,901. After the intervention (closing valuation which was October 2015) there were 440 pigs valued at USD 19,395 and by subtracting the opening valuation (value of pigs at the start of the year which was October 2014 in this case) from the closing valuation the herd value post-intervention was USD 5, 024. To obtain the gross margin from the pig enterprise, the variable costs were deducted from the livestock output. These consisted of pen repairs, veterinary drugs and feedstuff. Before the intervention, the total variable cost was USD 67, decreasing to USD 27 after the intervention so that the gross margin before the intervention was USD 6,697, rising to USD 17,556 after the intervention (a difference of USD 10,859). This is equivalent to USD 137 and USD 358 per household from the pig enterprise before and after the intervention respectively, so that the average gain was USD 221, increasing the household gross margin from pig rearing by a factor of 2.6. Table 5 provides a summary of the gross margin results. Apart from the determination of the gross margin from the pig enterprise, the study found that the total loss from pig condemnation by owners at the time of slaughter before intervention was USD 233, decreasing to USD 6 post intervention. Consequently, the total benefit across all the households from pig production was USD 11,086.

The gross margin from the pig enterprise was projected to the broader northern Lao PDR region by multiplying the equivalent total gross margin per household by the total number of pig-owning households in each province. The total gross margin for each province was then multiplied by 2.6 to estimate the total post intervention gross margin; resulting in a total gross margin of USD 9,398,720 before and USD 24,432 580 after the intervention as shown in Table 6; a projected difference of USD 15,033,860 at the regional level.

According to the village data, the annual post-intervention off-take was 24%, with 1.9% (1/ 51) slaughtered pigs condemned at home by their owners as the pork was grossly full of cysts. Projection of the same within four provinces in northern Lao revealed that of a total number of pigs slaughtered (84,488 or 1,934,775.2 kg of pork annually), an estimated 19,348 kg would be condemned, with a value of USD 58,043.

Item	Total income (USD)			
	Pre-intervention	Post intervention		
1. Pigs and pork products 'out'				
Total value of sold pigs	3,519	8,364		
Total value of slaughtered pigs	2,232	3,856		
Total value of pigs gifted/stolen/disappeared	768	1,239		
Subtotal: Value of pigs/pork products 'out'	6,519	13,459		
2. Pigs and pork products 'in'				
Total value of purchased pigs	960	630		
Total value of pigs received as gifts	737	231		
Subtotal: Value of pigs/pork products 'in'	1,697	860		
3. Change in herd value				
Closing valuation	14,372	19,395		
Opening valuation	12,471	14,372		
Overall change in herd value	1,901	5,024		
Livestock output	6,723	17,623		
4. Variable cost to farmers	26	67		
5. Total gross margin	6,697	17,556		
6. Average HH income from pig enterprise	137	358		

#### Table 5. Village pig gross margin analysis.

https://doi.org/10.1371/journal.pntd.0006782.t005

Consequently, the difference in the value of condemned pork before and after the intervention was estimated to be USD 23,990. Overall, the total benefit of controlling *T. solium* cysticercosis to the agricultural sector was USD 15,115,893, computed as the benefit of reduced pork condemnation (USD 23,990) plus increased regional household income (USD 15,057,850). It should be noted that the value of condemned pork was not part of the computation of the household income to avoid double counting of benefits.

# Extrapolated project cost

By estimating the human MDA coverage to be 63% [27] of 622,960 eligible participants over 4 years old, the total annual cost of MDA across all 4 Northern provinces was estimated to be USD 5,606,640. In the agricultural sector, the regional cost of treating pigs using TSOL18, CSF and OFZ was estimated to be USD 3,476,213, USD 3,125,013 and USD 3,722,053 respectively; totalling USD 10,323,279. Combining these figures resulted in a total project cost of USD 15,929,919 for the simultaneous control of *T. solium* taeniasis/cysticercosis, STH and CSF in the broader northern Lao PDR region.

Province	Number of pig rearing households	Total gross margin pre-intervention across all households (USD)	Total gross margin post-intervention across all households (USD)
Huaphan	28,900	2,586,550	6,725,030
Luang Prabang	26,900	2,975,140	7,733,750
Oudomxay	26,100	1,855,710	4,823,280
Phongsaly	22,800	1,981,320	5,150,520
Total	104, 700	9,423,600	24,432,580

Table 6. Annual pig gross margin analysis in northern Lao PDR.

https://doi.org/10.1371/journal.pntd.0006782.t006

# Determining the total cost-effectiveness of the intervention

By subtracting the livestock benefits (increased gross margin from pig enterprise and decreased value of condemned pork) and the averted cost in health expenditure from the project cost, the total cost from TSOL18, OFZ, CSF and human MDA was USD 833,785. Subsequently, the net cost-effectiveness of simultaneously controlling *T. solium* taeniasis/cysticercosis, CSF and STH was USD 14 per DALY averted. By computing the cost-effectiveness of *T. solium* taeniasis/cysticercosis control in the human population without integrating STH, and *T. solium* taeniasis/cysticercosis and STH (without addressing the pig population), the net cost-effectiveness of these approaches would be USD 1,609 and USD 93 per DALY averted respectively. By incorporating the pig intervention (TSOL18 and OFZ) with *T. solium* taeniasis/cysticercosis control only, the net cost-effectiveness was projected to be USD 3,672; by using the same regime but including STH in the calculations, the net DALY averted is USD 214. Table 7 provides a summary of the net cost per DALY averted for each of the *T. solium* taeniasis/cysticercosis control approaches.

By comparing the cost per DALY averted for each intervention approach with the Lao PDR GDP per capita as a measure of cost-effectiveness, it was found that the highly cost-effective approaches were i) control of T. solium taeniasis/cysticercosis in the human and pig populations, incorporating both CSF and STH control (14 USD/DALY averted) ii) control of T. solium taeniasis/cysticercosis and STH in the human population only (214 USD/DALY averted) and iii) control of T. solium taeniasis/cysticercosis in both the human and pig populations, incorporating STH control (93 USD/DALY averted). Control of T. solium taeniasis/cysticercosis only (1,609 USD/DALY averted) was found to be cost-effective while the least costeffective approach was incorporating the pig intervention (TSOL18 and oxfendazole) with T. solium cysticercosis control only (3,672 USD/DALY averted). Using zDALY approach, ALE for controlling T. solium taeniasis/cysticercosis was 13 and the HLE was 6. By adding DALY averted, which in this case was 3, 478 as estimated in Table 7, ALE and HLE, the T. solium taeniasis/cysticercosis zDALY was 3, 497. To compute the cost-effectiveness ratio of controlling T. solium taeniasis/cysticercosis in people and pigs without incorporating STH and CSF, the project cost (USD 5,606, 640 as estimated in Table 7) was divided by the zDALY yielding USD 3,662 per zDALY averted. Also, to compute the overall cost effectiveness ratio of combined control of all the three diseases (T. solium taeniasis/cysticercosis, STH and CSF), the ALE, HLE and DALY averted were totaled to determine the zDALY. Livestock benefit in terms of ALE was found to be 8,398, the HLE was 21 and the DALY averted was 59,556, thus the zDALY was 67,975 and the cost effectiveness ratio for the combined control of all the three diseases was

Table 7.	Net cost	t-effectiveness	of the	various	intervention	approaches.
I GOIC / I	1100 000	e enteetti enteoo	or the	1411040	miter vention	approacticos

Item	Intervention approach					
	Human intervention		Human and pig intervention			
	Control of <i>T. solium</i> only (A)	Control of <i>T. solium</i> + STH (B)	TSOL18, OFZ and (A)	TSOL18, OFZ and (B)	TSOL18, OFZ, CSF, (A) and (B)	
a) Project cost (USD)	5,606,640	5,606,640	12,804,906	12,804,906	15,929,919	
b) Livestock benefits (agricultural sector) (USD)	-	-	23,990	23,990	15,057,850	
c) Health expenditure averted (health sector) (USD)	10,234	38,284	10,234	38,284	38,284	
d) Sub-total ( <b>a</b> <i>minus</i> <b>b</b> <i>minus</i> <b>c</b> ) (USD)	5,596,406	5,568,356	12,770,682	12,742,632	833,785	
e) DALYs averted (health sector)	3,478	59,556	3,478	59,556	59,556	
f) Net cost per DALY averted ( <b>d/e</b> ) (USD/ DALY)	1,609	93	3,672	214	14	

https://doi.org/10.1371/journal.pntd.0006782.t007



**Fig 2. Sensitivity analysis of the neurocysticercosis associated DALY estimate for Lao.** Epilepsy is denoted as E and persons with epilepsy as PWE in the diagram.

https://doi.org/10.1371/journal.pntd.0006782.g002

USD 234 per zDALY averted; representing 13% of Laos per capita GDP and this was well within the WHO's threshold of very cost-effective interventions.

#### Sensitivity analysis

Sensitivity analysis using partial correlation coefficients showed that prevalence, mortality rate and disability weights were very influential in the disease burden models. For example, the prevalence of NCC, mortality rate, disability weight of the untreated, prevalence rate of epilepsy and disability weight of the epilepsy cases treated were the most influential in modelling the burden of *T. solium* taeniasis/cysticercosis as shown in Fig 2.

# Discussion

Although methods to estimate the costs of zoonotic disease to both livestock productivity and humans have been proposed [68-69], there has been no totally satisfactory conceptual framework for analyzing the total societal burden of zoonotic disease; that is the combined costs of disease from both the humans and animals [70]. The recently developed concept of the zDALY addresses this gap [65]. This study had both zoonotic (T. solium cysticercosis) and non-zoonotic (STH and CSF) diseases. T. solium cysticercosis as a zoonotic disease had YLL, YLD, HLE and ALE components of burden of disease, while STH as a non-zoonotic disease had YLL, YLD and HLE components. Given CSF is an animal disease it only had the ALE metric. Therefore based on the YLL, YLD, HLE and ALE disease burden metrics both aggregate net cost and zDALY approaches were used to estimate the changes in cost effectiveness ratio when externalities generated by treatment of T. solium taeniasis/cysticercosis are included. The aggregate net method revealed that the 'highly cost-effective' approach for northern region of Lao PDR is the combined human-animal approach incorporating T. solium taeniasis/cysticercosis control with two additional diseases; STH and CSF control (USD 14 per DALY averted) and zDALY approach corroborated this finding given the overall cost effectiveness ration of controlling all the three diseases (T. solium taeniasis/cysticercosis, STH and CSF) was USD 234 per zDALY averted representing 13% of the GDP per capita falling well within WHO's threshold of very cost-effective interventions. Other cost-effective approaches included the human MDA intervention targeting both T. solium taeniasis/cysticercosis and STH (USD 93 per DALY averted), and the combined human-pig intervention that targeted both T. solium taeniasis/cysticercosis and STH (USD 214 per DALY averted). The least costeffective intervention approaches were those that addressed *T. solium* taeniasis/cysticercosis in isolation, regardless of whether this was in the human population (USD 1,609 per DALY averted), or jointly with an intervention in the pigs (USD 3,672 per DALY averted). Consequently, the results show that control of *T. solium* taeniasis/cysticercosis alone in humans and pigs is not cost-effective whereas control of STH in humans is. Also, the results obtained from using zDALY approach confirmed that it is not cost-effective to control *T. solium* taeniasis/ cysticercosis alone in humans and pigs without incorporating STH and CSF in northern Lao; zDALY metric was very close to the findings from the aggregate net cost method. Also, the null hypothesis was rejected given that addition of STH and CSF markedly improved the overall cost effectiveness of controlling *T. solium* taeniasis/cysticercosis. Therefore, this study concluded that integrating *T. solium* taeniasis/cysticercosis control with other cost-effective programmes is recommended to effectively control it in Laos.

Information obtained from the semi-structured questionnaire supported previous findings that revealed smallholder pig rearing to be an important farming activity in the study area as also revealed by Bardosh et al (2014) [71]. The intervention resulted in improved pig productivity, seen as an increase in the average number of pigs reared per household from 5.2 to 8.4 after 12 months of the intervention, and a reduction in pre-weaning mortality from 48.3% to 8.4% due to CSF vaccination; low mortality was probably the main course of increased number of growers as most piglets were surviving and reaching this age. Apart from CSF vaccination, deworming of pigs (especially free ranging ones) with OFZ potentially played a role in improving the overall cost effectiveness of the intervention by protecting pigs from new T. solium cysticercosis infections thus protecting humans; as well as improving the health of pigs as it has an effect on nematode infections which are a source of disease and production losses. These results corresponded to the increased gross margin from the pig enterprise; remarkably the greatest production age increase was seen in growers, from 12% to 44% of the overall herd composition, highlighting the importance of integrating disease interventions into future pig productivity improvement projects. In the study area, the combination of animal health interventions with the availability of improved feeding, which had been established prior to the intervention, allowed farmers to take full advantage of the production capacity of their livestock, once animal health had been restored. Although the human health benefits alone fully justify the investment as demonstrated through the economic impact of averted DALYs, the combination of such an intervention with improved production approaches adds considerable value to the overall intervention. It might also give an additional incentive to farmers if the effect is large enough for them to notice the production-and in consequence economic-benefit. Also, vaccinating pigs with TSOL18 ensured a lifetime immunity to T. solium cysticercosis for pigs, reducing the risk of acquiring infection long term.

This study has shown that the inclusion of approaches that are effective against pig production diseases such as CSF has played a major role in increasing the cost-effectiveness in regions where *T. solium* cysticercosis and CSF are co-endemic. To achieve high cost-effectiveness in future, pig vaccination against *T. solium* cysticercosis could be done together with CSF or an equivalent bivalent 'One Health' vaccine developed for regions where CSF is endemic; *T. solium* cysticercosis does not typically affect pig productivity, it will be difficult to convince farmers to pay for *T. solium* cysticercosis vaccine unless they are likely to be penalized for cystic pork. Equally, where meat inspection practices are not well managed, *T. solium* cysticercosis interventions should focus on diseases or management practices that decrease pig mortality (pre-weaning mortality in particular), so as to achieve a higher survival rate and thus increase the overall livestock productivity benefits of the *T. solium* cysticercosis intervention.

There is a need for sharing resources between agricultural and health sectors, especially where the inclusion of secondary diseases such as STH and CSF play a major role in the

benefits accrued to each sector. Joint disease control is a critical component of enhancing household health, wealth and overall wellbeing, given the biggest beneficiaries are the affected households. Unfortunately, integrated sectoral approaches under the One Health movement are rare, despite an acknowledged need to tackle societal problems such as zNZDs in a comprehensive manner [72]. A major reason observed for the lack of sustainable One Health approaches in veterinary public health is related to the concept of who should fund what, particularly where cost sharing between sectors is expected. However, this study clearly demonstrates that integrated actions at a larger scale are significantly more cost effective than 'vertical' disease approaches that address issues individually, and thus should be the guiding principle for addressing future *T. solium* taeniasis/cysticercosis interventions, or those against the zNTDs more generally.

This study has limitations, the primary observation being the large amount of secondary data used to simulate and estimate the cost-effectiveness of controlling T. solium taeniasis/cysticercosis, STH and CSF in the northern Lao region after assuming a linear scaling out of the intervention. The use of significant secondary data sources is not without precedent for estimations of T. solium taeniasis/cysticercosis burden [73-74], and highlights the current dearth of data globally for this disease resulting from and contributing to its neglected status. Further studies are needed to establish the T. solium taeniasis/cysticercosis, STH and CSF prevalent regions in northern Lao PDR, or indeed the broader southeast Asia region more generally; it would be prudent to focus initially on potential hyper-endemic T. solium taeniasis/cysticercosis 'hotspots', identified by a combination approach of both social and epidemiological methods. A second limitation, when looking at the aggregated societal benefits and the net monetary benefit, high livestock benefits may mean that monetary benefits exceed monetary costs. This would show that the programme is dominant: effective and cost-saving; in this situation, calculating incremental cost-effectiveness ratio is not relevant. This is a difficult result to interpret, or rank, and could have the unwanted effect of skewing the allocation of cost entirely towards the livestock sector, since livestock benefits outweigh costs. This would be a particularly unhelpful outcome, as the strength of this intervention is that it simultaneously deals with both the human and livestock disease reservoirs, resulting in greater sustainability. However other methodologies such as zDALY can be used to estimate monetary losses in livestock which can then be incorporated into the DALY estimate particularly if the intervention only involves zoonotic diseases. Other limitations include use of the village data with small sample size, use of data from a hyper-endemic foci to project the cost effectiveness of the intervention and lack of definitive diagnosis of NCC. Consequently the attribution of NCC to epilepsy in northern Lao PDR might be lower than stated in this study and further studies are needed to find out if this is the case and whether more T. solium taeniasis/cysticercosis hyper-endemic foci exist in northern Lao PDR. However, information obtained from the study area coupled with the sensitivity analysis on the assumptions used to estimate the DALY provides a good basis of understanding the impact of simultaneously controlling T. solium taeniasis/cysticercosis, STH and CSF.

# Conclusion

Control of *T. solium* taeniasis/cysticercosis in the northern Lao PDR is currently heavily dependent on therapeutic interventions in the human or pig populations–ideally both–to reduce the disease prevalence. However, sustainable control of *T. solium* taeniasis/cysticercosis should not be taken in isolation; incorporating the control of other pig production diseases (such as CSF) and/or soil transmitted helminth control is recommended to maximize the intervention cost-effectiveness. This is especially true for interventions that the farmer is

expected to pay for; incorporating production-impacting diseases into *T. solium* taeniasis/cysticercosis control will incentivize farmers to pay for its control. This cost-effectiveness analysis clearly shows that controlling *T. solium* taeniasis/cysticercosis in isolation is not cost effective, and more holistic, innovative methods to build zNTD control into existing human health or livestock production and development programmes would be beneficial.

# Supporting information

**S1 Checklist. STROBE checklist.** (DOC)

**S2 Checklist. CHEERS checklist.** (DOC)

# Acknowledgments

The project would like to thank the Lao PDR Ministries of Health and Agriculture for their support and assistance in undertaking the field work; in particular the staff from the Mai District and Phongsaly Provincial Health and Agricultural Offices.

## **Author Contributions**

**Conceptualization:** Walter O. Okello, Anna L. Okello, Phouth Inthavong, Tassilo Tiemann, Brecht Devleesschauwer, Alexandra Shaw, John Allen.

Data curation: Walter O. Okello, Brecht Devleesschauwer.

Formal analysis: Walter O. Okello, Brecht Devleesschauwer, Alexandra Shaw.

Funding acquisition: Tassilo Tiemann, John Allen.

**Investigation:** Walter O. Okello, Phouth Inthavong, Tassilo Tiemann, Ammaly Phengsivalouk.

Methodology: Walter O. Okello, Brecht Devleesschauwer, Alexandra Shaw.

Project administration: Anna L. Okello, Phouth Inthavong, Tassilo Tiemann, John Allen.

Resources: Ammaly Phengsivalouk, John Allen.

Supervision: Alexandra Shaw, John Allen.

Validation: Walter O. Okello, Brecht Devleesschauwer, John Allen.

Visualization: Walter O. Okello.

Writing – original draft: Walter O. Okello, Anna L. Okello, Phouth Inthavong, Tassilo Tiemann, Ammaly Phengsivalouk, Brecht Devleesschauwer, Alexandra Shaw, John Allen.

Writing – review & editing: Walter O. Okello, Anna L. Okello, Tassilo Tiemann, Ammaly Phengsivalouk, Brecht Devleesschauwer, Alexandra Shaw, John Allen.

# References

- Schantz P. Taenia solium cysticercosis: an overview of global distribution and transmission. In: Singh G and Prabhakar S, editors. Taenia solium Cysticercosis from Basic to Clinical Science. CABI Publishing, Wallingford; 2002. pp. 63–73.
- 2. Newton CR and Garcia HH. Epilepsy in poor regions of the world. Lancet. 2010; 380: 1193–1201

- Torgerson PR, Devleesschauwer B, Praet N, Speybroeck N, Willingham AL, Kasuga F, et al. World Health Organization Estimates of the Global and Regional Disease Burden of 11 Foodborne Parasitic Diseases, 2010: A Data Synthesis. PLoS Med. 2015; 12(12): e1001920. <u>https://doi.org/10.1371/journal.pmed.1001920</u> PMID: 26633705
- 4. Fenwick A. The global burden of neglected tropical diseases. Public Health. 2012; 125: 233–236.
- Willingham AL and Engels D. Control of *Taenia solium* cysticercosis/taeniosis. Adv Parasitol. 2006; 61: 509–566. https://doi.org/10.1016/S0065-308X(05)61012-3 PMID: 16735172
- Zinsstag J, Schelling E, Roth F, Bonfoh B, de Savigny D and Tanner M. Human benefits of animal interventions for zoonosis control. Emerg Infect Dis. 2007; 13(4):527–531. https://doi.org/10.3201/eid1304. 060381 PMID: 17553265
- Atun R, de Jongh T, Secci F, Ohiri K and Adeyi O. Integration of targeted health interventions into health systems: a conceptual framework for analysis. Health Policy and Plan. 2010; 25: 104–111.
- Gre ´pin KA and Reich MR. Conceptualizing Integration: A Framework for Analysis Applied to Neglected Tropical Disease Control Partnerships. PLoS Negl Trop Dis. 2008; 2 (4): e174. https://doi.org/10.1371/ journal.pntd.0000174 PMID: 18446203
- 9. Shaw A. 2009. The Economics of Zoonoses and Their Control. In: Rushton J, editor. The Economics of Animal Health and Production. London: CABI; 2009.161–167.
- Mecaskey JW, Ngirwamungu E and Kilima PM. Integration of trachoma control into primary health care: the Tanzanian experience. 2003. Am J Trop Med Hyg; 69(5): 29–32.
- 11. Utzinger J and Keiser J. Schistosomiasis and soil-transmitted helminthiasis: common drugs for treatment and control. Exp Opin pharmacother. 2004; 5: 263–285.
- Cleaveland S, Fèvre EM, Kaare M and Coleman PG. Estimating human rabies mortality in the United Republic of Tanzania from dog bite injuries. Bull World Health Organ. 2002; 80(4): 304–310. PMID: 12075367
- Zinsstag J, Schelling E, Wyss K, Bechir M. Potential of cooperation between human and animal health to strengthen health systems. Lancet. 2005; 366: 2142–2145. https://doi.org/10.1016/S0140-6736(05) 67731-8 PMID: 16360795
- Molyneux DH, Hotez PJ and Fenwick A. Rapid-Impact Interventions: How a Policy of Integrated Control for Africa's Neglected Tropical Diseases Could Benefit the Poor. PLoS Med 2(11). 2005: e336. https:// doi.org/10.1371/journal.pmed.0020336 PMID: 16212468
- 15. Utzinger J and de Savigny D. Control of neglected tropical diseases: integrated chemotherapy and beyond. PLoS Med. 2006; 3: e112. https://doi.org/10.1371/journal.pmed.0030112 PMID: 16435910
- Roth F, Zinsstag J, Orkhon D, Chimed-Ochir G, Hutton G, Cosivi O, Carrin G and Otte J. Human health benefits from livestock vaccination for brucellosis: case study. Bull World Health Organ. 2003; 81 (12): 867–876. PMID: 14997239
- Okello A, Thomas L, Inthavong P, Ash A, Khamlome B, Keokamphet C, Newberry K, Gauci CG, Gabriël S, Dorny P, RC Thompson ARC, Lightowlers MW and Allen J. Assessing the impact of a joint human-porcine intervention package for *Taenia solium* control: Results of a pilot study from northern Lao PDR. Acta Trop. 2016; 159: 185–191. <u>https://doi.org/10.1016/j.actatropica.2016.03.012</u> PMID: 26992295
- Tran DS, Odermatt P, Le TO, Huc P, Cabanac MD, Barennes H, Strobel M and Preux PM. Prevalence of Epilepsy in a Rural District of Central Lao PDR. Neuroepidemiology. 2006; 26:199–206. https://doi. org/10.1159/000092407 PMID: 16569936
- Tran DS, Zen J, Strobel M, Odermatt P, Preux PM, Huc Pierre, Delneuville L and Barennes H. The challenges of epilepsy control in deprived settings: Low compliance and high fatality rates during a community-based Phenobarbital program in rural Lao. Epilepsia. 2008; 49(3).
- Jeon HK, Yong TS, Sohn WM, Chai JY, Min MY, Rim HJ, Insisiengmay B and Eom KS. 2013. Human Neurocysticercosis Case and an Endemic Focus of *Taenia solium* in Lao. Korean J of Parasitol. 2013; 51 (5): 599–602.
- Tran DS, Odermatt P, Singphuoangphet S, Cabanac MD, Preux PM, Strobel M and Barennes H. Epilepsy in Lao PDR: Knowledge, attitudes, and practices in the community. Epilepsy Behav. 2007; 10: 565–570 https://doi.org/10.1016/j.yebeh.2007.02.018 PMID: 17446140
- Harimanana A, Chivorakul P, Souvong V, Preux PM and Barennes H. Is insufficient knowledge of epilepsy the reason for low levels of healthcare in the Lao PDR? BMC Health Serv Res. 2013; 13:41. https://doi.org/10.1186/1472-6963-13-41 PMID: 23379720
- Kyvsgaard NC, Johansen MV and Carabin H. Simulating transmission and control of Taenia solium infections using a Reed-Frost stochastic model. Int J Parasitol. 2007; 37(5): 547–558. https://doi.org/ 10.1016/j.ijpara.2006.11.018 PMID: 17250839

- Thomas LF. Landscape Analysis: Control of Taenia solium. Geneva, Switzerland: World Health Organization. 2015. Available from: http://apps.who.int/iris/bitstream/10665/164359/1/9789241508643\_eng. pdf
- Garcia HH, Gonzalez AE, Tsang VCW, O'neal SE, Llanos–Zavalaga F, Gonzalvez G, Romero J, Rodriguez S, Moyano LM, Ayvar V, Diaz A, Hightower A, Craig PS, Lightowlers MW, Gauci CG, Leontsini E and Gilman RH. Elimination of Taenia solium Transmission in Northern Peru. N Engl J Med. 2016; 374 (24): 2335–2344. https://doi.org/10.1056/NEJMoa1515520 PMID: 27305193
- 26. Phommasack B, Saklokham K, Chanthavisouk C, Nakhonesid-Fish V, Strandgaard H, et al. Coverage and costs of a school deworming programme in 2007 targeting all primary schools in Lao PDR. Trans R Soc Trop Med Hyg. 2008; 102(12): https://doi.org/10.1016/j.trstmh.2008.04.036
- Ash A, Okello A, Khamlome B, Inthavong P, Allen J and Thompson RAC. Controlling *Taenia solium* and soil transmitted helminths in a northern Lao PDR village: Impact of a triple dose albendazole regime. Acta Trop. 2015. Available from: http://dx.doi.org/10.1016/j.actatropica.2015.05.018
- Lightowlers MW. Control of Taenia solium taeniasis/cysticercosis: past practices and new possibilities. Parasitology. 2013; 140: 1566–1577. https://doi.org/10.1017/S0031182013001005 PMID: 23947762
- Pondja A, Neves L, Mlangwa J, Afonso S, Fafetine J, Willingham AL III, et al. Use of Oxfendazole to Control Porcine Cysticercosis in a High-Endemic Area of Mozambique. PLoS Negl Trop Dis. 2012; 6 (5): e1651. https://doi.org/10.1371/journal.pntd.0001651 PMID: 22666509
- Blacksell SD, Khounsy S, Boyle DB, Gleeson LJ, Westbury HA and Mackenzie JS. Genetic typing of classical swine fever viruses from Lao PDR by analysis of the 5' non-coding region. Virus Genes. 2005; 31(3):349–355. https://doi.org/10.1007/s11262-005-3253-0 PMID: 16175340
- Blacksell SD, Khounsy S, Boyle DB, Greiser-Wilke I, Gleeson LJ, Westbury HA and Mackenzie JS. Phylogenetic analysis of the E2 gene of classical swine fever viruses from Lao PDR. Virus Res. 2004; 104(1): 87–92. https://doi.org/10.1016/j.virusres.2004.02.041 PMID: 15177896
- Blacksell SD, Khounsy S, Boyle DB, Van Aken D, Gleeson LJ and Westbury HA. Comparative susceptibility of indigenous and improved pig breeds to Classical swine fever infection: Practical and epidemiological implications in a subsistence-based, developing country setting. Trop Anim Health Prod. 2006; 38: 467–474. PMID: 17243474
- Okello AL, Ash A, Keokhamphet C, Hobbs E, Khamlome B, Dorny P, Thomas L and Allen J. Investigating a hyper-endemic focus of *Taenia solium* in northern Lao PDR. Parasit Vectors. 2014; 7:134 <a href="https://doi.org/10.1186/1756-3305-7-134">https://doi.org/10.1186/1756-3305-7-134</a> PMID: 24678662
- 34. Kashinaga M. The transmission of written genealogies and patrilineality among the Tai Dam. Senri Ethnol. Stud. 2009; 74, 97e115.
- International League Against Epilepsy. Guidelines for Epidemiologic Studies on Epilepsy. Epilepsia. 1993; 34(4):592–596. PMID: 8330566
- Placencia M, Sander JW, Shorvon SD, Ellison RH, Cascante SM. Validation of a screening questionnaire for the detection of epileptic seizures in epidemiological studies. Brain J Neurol. 1992; 115:783– 94.
- Drummond MF, Sculpher MJ, Torrance GW, O'Brien BJ and Stoddart GL. Methods for the economic evaluation of healthcare programmes, third edition. Oxford: Oxford University Press, 2005.
- 38. Reynolds IN. Selecting the proper depreciation method. Account Review. 1961; 36(2): 239–248.
- **39.** Delignette-Muller ML and Dutang C. fitdistrplus: An R Package for Fitting Distributions. J Stat Softw. 2015; 64(4): 1–34.
- Rushby FJA and Hanison K. Calculating and presenting disability adjusted life years (DALYs) in costeffectiveness analysis. Health Policy and Plan. 2001; 16(3): 326–331.
- **41.** Murray CJL and Lopez AD. The global burden of disease, 1990–2020. Nat Med. 1998; 14(11): 1241– 1243.
- R Core Team. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. 2008. Available from: http://www.R-project.org/.
- Devleesschauwer B. The socioeconomic impact of Taenia solium cysticercosis. github 2017: <u>https://github.com/brechtdv/tsol-mozambique</u>
- 44. Preux PM and Cabanac MD. Epidemiology and aetiology of epilepsy in sub-Saharan Africa. Lancet Neurol. 2005; 4: 21–31. https://doi.org/10.1016/S1474-4422(04)00963-9 PMID: 15620854
- 45. Devleesschauwer B, McDonald S, Haagsma J, Praet N, Havelaar A and Speybroeck N. DALY: The DALY Calculator—A GUI for stochastic DALY calculation in R. R package version 1.2.0. 2013. Available from: http://cran.rproject.org/package=DALY
- 46. Conlan JV, Khamlome B, Vongxay K, Elliot A, Pallant L, Sripa B, Blacksell SD, Fenwick S and Thompson RCA. Soil-Transmitted Helminthiasis in Lao PDR: A Community-Wide Cross-Sectional Study of

Humans and Dogs in a Mass Drug Administration Environment. Am J Trop Med Hyg. 2012; 86(4): 624–634. https://doi.org/10.4269/ajtmh.2012.11-0413 PMID: 22492147

- **47.** Eom KS, Yong TS, Sohn WM, Chai JY, Min DY, Rim HJ, Jeon HK, Banouvong V, Insisiengmay B and Phommasack B. Prevalence of Helminthic Infections among Inhabitants of Lao PDR. Korean J of Parasitol. 2014; 52(1): 51–56.
- **48.** Schroder SA. Incidence, prevalence, and hybrid approaches to calculating disability-adjusted life years. Popul Health Metr. 2012; 10:19. https://doi.org/10.1186/1478-7954-10-19 PMID: 22967055
- 49. Devleesschauwer B, Havelaar AH, de Noordhout CM, Haagsma JA, Praet N, Dorny P, Duchateau L, Torgerson PR, Oyen HV and Speybroeck N. DALY calculation in practice: a stepwise approach. Int J Public Health. 2014; 59(3): 571–574. https://doi.org/10.1007/s00038-014-0553-y PMID: 24748107
- 50. Lao Statistics Bureau. Population Census 2005. 2005; Available from: http://lsb.gov.la
- Carabin H, Krecek RC, Cowan LD, Michael L, Foyaca-Sibat H, Nash T and Willingham AL. Estimation of the cost of Taenia Solium cysticercosis in Eastern Cape Province, South Africa. Trop Med Int Health; 11(6): 906–916. https://doi.org/10.1111/j.1365-3156.2006.01627.x PMID: 16772013
- Rajshekhar V, Joshi DD, Doanh NQ, van De N and Xiaonong Z. Taenia solium taeniosis/cysticercosis in Asia: epidemiology, impact and issues. Acta Trop. 2003; 87: 53–60. PMID: 12781378
- Ding D, Hong Z, Wang W, Wu JZ, De Boer HM, Prilipko L and Sander JW. Assessing the Disease Burden Due to Epilepsy by Disability Adjusted Life Year in Rural China. Epilepsia. 2006; 47(12):2032– 2037. https://doi.org/10.1111/j.1528-1167.2006.00802.x PMID: 17201700
- 54. United Nations Statistics Division. 2015. Lao PDR Population by age, sex and urban/rural residence. Available from: http://data.un.org/Data.aspx?d=POP&f=tableCode%3a22
- 55. Murray CJL and Lopez AD. The global burden of disease: a comprehensive assessment of mortality and disability from diseases, injuries and risk factors in 1990 and projected to 2020. 1996. Cambridge: Harvard University Press.
- 56. Salomon JA, Haagsma JA, Davis A, et al. Disability weights for the Global Burden of Disease 2013 study. Lancet Glob Health 2015; 3: e712–23. https://doi.org/10.1016/S2214-109X(15)00069-8 PMID: 26475018
- Hotez PJ, Alvarado M, Basa ´n ez MG, Bolliger I, Bourne R, et al. The Global Burden of Disease Study 2010: Interpretation and Implications for the Neglected Tropical Diseases. PLoS Negl Trop Dis. 2014; 8 (7): e2865. https://doi.org/10.1371/journal.pntd.0002865 PMID: 25058013
- Bundy DAP, Chan MS, Medley GF, Jamison D and Savioli L. Intestinal Nematode Infections. In: Murray CJL, Lopez AD, Mathers CD, editors. Global Epidemiology of Infectious Disease. WHO: 243–300. Global Burden of Disease Volume IV. 2004. Available from: http://whqlibdoc.who.int/publications/2004/ 9241592303.pdf
- Pullan RL, Smith JL, Jasrasaria R and Baker S. Global numbers of infection and disease burden of soil transmitted helminth infections in 2010. Parasit Vectors. 2014; 7:37 <u>https://doi.org/10.1186/1756-3305-7-37 PMID: 24447578</u>
- Rushton J, Thornton PK and Otte MJ. Methods of economic impact assessment. Rev sci tech Off int Epiz. 1999; 18 (2): 315–342.
- 61. Lao Statistics Bureau. Lao Agricultural Census 2010–2011. 2011; Available from: http://lsb.gov.la
- Mathers CD, Ezzati M and Lopez AD. Measuring the Burden of Neglected Tropical Diseases: The Global Burden of Disease Framework. PLoS Negl Trop Dis. 2007 1(2): e114. <u>https://doi.org/10.1371/journal.pntd.0000114 PMID: 18060077</u>
- Cohen DJ and Reynolds MR. Interpreting the Results of Cost-Effectiveness Studies. J Am Coll Cardiol. 2008; 52(25): 2119–2126. https://doi.org/10.1016/j.jacc.2008.09.018 PMID: 19095128
- 64. Shaw APM, Rushton J, Roth F and Torgerson PR. DALYs, dollars and dogs: how best to analyse the economics of controlling zoonoses. Rev. Sci. Tech. Off. Int. Epiz. 2017; 36 (1): 147–161.
- 65. Torgerson PR, Rüegg S, Devleesschauwer B, Abela-Ridder B, Havelaar AH, Shaw APM, Rushton J and Speybroeck N. zDALY: An adjusted indicator to estimate the burden of zoonotic diseases. One Health. 2018; 5: 40–45. https://doi.org/10.1016/j.onehlt.2017.11.003 PMID: 29911164
- World Health Organization (WHO). Cost-effectiveness and strategic planning (WHO-CHOICE). Available from: <a href="http://www.who.int/choice/costs/CER\_thresholds/en/">http://www.who.int/choice/costs/CER\_thresholds/en/</a>
- 67. World Bank. GDP data. Available from: <u>https://data.worldbank.org/indicator/NY.GDP.PCAP.KD?</u> locations=IN
- Tschopp R, Hatterndorf J, Roth F, Choudhoury A, Shaw A, Aseffa A and Zinsstag J. Cost Estimate of Bovine Tuberculosis to Ethiopia. In: Mackenzie JS, Jeggo M, Daszak P and Richt JA, editors. One Health: The Human-Animal-environment Interfaces in Emerging Infectious Diseases: Springer-Verlag Berlin Heideberg; 2013. pp 249–268. https://doi.org/10.1007/978-3-642-36889-9

- Narrod C, Zinsstag J and Tiongco M. A One Health Framework for estimating the economic costs of zoonotic diseases on Society. EcoHealth. 2012; https://doi.org/10.1007/s10393-01
- 70. Häsler B, Cornelsen L, Bennani H and Rushton J. A review of the metrics for One Health benefits. Rev. Sci. Tech. Off. Int. Epiz. 2014; 33 (3): 453–464.
- Bardosh K, Inthavong P, Xayaheuang S and Okello AL. Controlling parasites, understanding practices: The biosocial complexity of a One Health intervention for neglected zoonotic helminths in northern Lao PDR. Soc Sci Med. 2014; 120: 215–223. https://doi.org/10.1016/j.socscimed.2014.09.030 PMID: 25261615
- 72. One health: the theory and practice of integrated health approaches. Edited by Zinsstag J, Schelling E, David Waltner-Toews D, Whittaker M and Tanner M. 2015. CAB International. ISBN-13: 978 1 78064 341 0
- **73.** Praet N, Speybroeck N, Manzanedo R, Berkvens D, Nsame Nforninwe D, et al. The Disease Burden of Taenia solium Cysticercosis in Cameroon. PLoS Negl Trop Dis. 2009: 3(3): e406. <u>https://doi.org/10.1371/journal.pntd.0000406</u> PMID: 19333365
- 74. Bhattarai R, Budke CM, Carabin H, Proan<sup>o</sup> JV, Flores-Rivera J, et al. Estimating the Non-Monetary Burden of Neurocysticercosis in Mexico. PLoS Negl Trop Dis. 2012; 6(2): e1521. https://doi.org/10. 1371/journal.pntd.0001521 PMID: 22363827